Author's response to reviews

Title: Common NOD2/CARD15 variants are not associated with susceptibility or the clinicopathologic characteristics of sporadic colorectal cancer in Hungarian patients

Authors:

Peter Laszlo Lakatos (kislakpet@yahoo.com)
Erika Hitre (hitre@oncol.hu)
Ferenc Szalay (szalay@bel1.sote.hu)
Kerstin Zinober (kislakpet@yahoo.com)
Peter Fuszek (fuszpet@bel1.sote.hu)
Laszlo Lakatos (lakatos.laszlo@vmkorhaz.hu)
Simon Fischer (sfischer712@yahoo.ca)
Janos Osztovits (osztoka@yahoo.co.uk)
Gemela Orsolya (gemiorsl@freemail.hu)
Gabor Veres (vergab@gyer1.sote.hu)
Janos Papp (papjan@bel1.sote.hu)
Peter Ferenci (peter.ferenci@meduniwien.ac.at)

Version: 4 Date: 22 March 2007

Author's response to reviews:

22nd of March 2007

Melissa Norton, MD
Medical Editor
BMC Cancer
http://www.biomedcentral.com/bmccancer

Dear Editor,

Please find attached our revised manuscript:

Common NOD2/CARD15 variants are not associated with susceptibility or the clinicopathologic characteristics of sporadic colorectal cancer in Hungarian patients

Authors: Peter Laszlo Lakatos, Erika Hitre, Ferenc Szalay, Kerstin Zinober, Peter Fuszek, Laszlo Lakatos, Simon Fischer, Janos Osztovits, Orsolya Gemela, Gabor Veres, Janos Papp, Peter Ferenci

First of all we would like to thank the reviewers for their careful re-assessment of our manuscript and the helpful comments.

Answers to Pia Alhopuro

Minor essential revisions:
1)Abstract: SNP8 R702W (not R703W) was corrected
2)Patients and Methods: The authors should specify the ethnicity of the patients in the control group. Caucasian ethnicity was included
3)Patients and Methods: detection of SNP8 is missing: SNP8 detection is included
4)Discussion: (page 8) <50 years (not >50 years) was corrected
5)Discussion: (page 9) In the study by Roberts et al, the association of G908R and 3020insC with CRC were not statistically significant. Text was modified as suggested

Formatting changes were done as requested.

We would like to thank you for accepting and publishing our paper. We do hope that the changes that have been made, have improved the quality of the manuscript also with regards to grammar.
Sincerely yours,

Laszlo Peter Lakatos, MD, PhD  
1st Department of Medicine  
Semmelweis University  
Koranyi str. 2/A, Budapest, H-1083 Hungary  
Tel: +36-1-210-0278 / 1500, 1520  
Fax: +36-1-313-0250  
e-mail: kislakpet@bel1.sote.hu